The Social Gradient in the Impact of the Chernobyl Accident: The Case of Austria^{*}

MARTIN HALLA

University of Linz & IZA

MARTINA ZWEIMÜLLER

University of Linz

incomplete draft paper October 31, 2012

^{*}Corresponding author: Martin Halla, Johannes Kepler University of Linz, Department of Economics, Altenbergerstr. 69, 4040 Linz, ph.: +437024688706, fax: +4370246828706, email: martin.halla@jku.at. For helpful discussion and comments, we would like to thank Anders Björklund, Daniel S. Hamermesh, Matthew Lindquist, Torsten Santavirta, and Rudolf Winter-Ebmer, as well as seminar participants at Vienna (University of Economics and Business, University of Technology), Stockholm (SOFI), Hamburg, Innsbruck, Nürnberg (IAB). The usual disclaimer applies. Moreover, we thank the following institutions for providing us with data: *Environment Agency Austria, Statistics Austria, Main Association of Austrian Social Security Institutions, Upper Austrian Health Insurance Fund*, and the *Austrian Federal Ministry of Finance*. This research was funded by the Körner Fonds and the Austrian FWF (NRN Labor Economics and the Welfare State).

1 Introduction

The importance of early-life conditions for outcomes in later life is now widely recognized among scholars in different disciplines. Economists are particularly interested in the effect of early events on the accumulation of human capital (Almond and Currie, 2011). However, several factors complicate the isolation of causal effects in this context. Ideally, one would observe a link between an early exogenous shock and later outcomes in a large-scale micro data set. For instance, Almond, Edlund and Palme (2009) interpret the radioactive fallout from the Chernobyl accident in 1986 as an early shock to study the role of latent variables in human capital formation. They find that Swedish children in low-educated families born in fall 1986 that where prenatally exposed to radioactive fallout had significantly lower grades in compulsory school at the age of 16. In contrast, for children in highly-educated families no comparable effect is observable. For neither group they detect any corresponding health damage. These estimates are identified by the difference in rainfall levels while the radioactive plume was over Sweden; which led to stark geographic variation in the levels of radioactive fallout.¹

While this research design is in principle very appealing the estimated long-term effects are still hard to interpret. The estimated effects may not only entail the biological effect of the shock, but also the parental response to it. Parents may either make compensating or reinforcing investments in the child's human capital. In the presence of asymmetric parental investments along different dimensions of human capital, such reduced-form estimates cannot be even unambiguously interpreted as upper or lower bounds of the biological effect (Conti, Heckman, Yi and Zhang, 2011). Thus, in order to fully understand the effect of early-life conditions for outcomes in later life, it is crucial to examine also the parental responses. In the case of the social gradient in the impact of the Chernobyl accident it is possible that highly-educated parents had a larger compensatory (or smaller reinforcing) response compared to low-educated parents.

In this paper we study the short and long-run effects of the Chernobyl accident on the Austrian fall 1986 cohort with a special focus on the response behavior of treated parents.² We aim to contribute to a better understanding of the reduced form estimates of early childhood events on later outcomes, and to uncover the sources of social gradients in these effects. Therefore, we extend the research design of Almond, Edlund and Palme (2009) along several important dimensions.

¹Whether radioactive fallout from the Chernobyl accident in 1986 had detrimental effects on individuals living in Western European countries or not is still a controversial question. While this is an inherently medical question, the clean identification strategy of Almond, Edlund and Palme (2009) distinguishes this paper from earlier studies (summarized in Appendix A).

²Austria ranks among the countries that received the most radioactive fallout. Differences in rainfall immediately after the accident caused substantial geographic variation in ground deposition of Caesium-137 fallout (half-life of 30 years) with maximum values of nearly 200 kilobecquerels per square meter. Only Russia, Ukraine, Belarus and some parts of Scandinavia had fallout values higher than 200 kilobecquerels per square meter; see Figure 3.5. in IAEA (2006).

First, we pay special attention to the tension between so-called *culling* and *scarring* effects.³ In particular, we provide empirical strategies to check whether radioactive exposure led to early (fetal) death, and whether these effects vary across low and highly-educated families. In the presence of culling, estimated effects (on the surviving population) may underestimate the true impact of the early shock. Second, we do not only consider prenatal radiation effects (i.e. the true causal effect of radioactive fallout), but we also aim to quantify non-radiation effects resulting from prenatal behavioral adjustment.⁴ Potential parents in treated but also in control regions—both unaware of the actual local level of radioactive exposure — may have changed their behavior in response to the accident. This may generate a selected sample of children conceived and/or born, as well as directly affect live births of this cohort. Again, this early response may differ across educational groups. Third, we re-examine the impact of the radioactive exposure (and the accident more general) on children's health at birth and later in life. Fourth, we examine the postnatal responsive investment behavior of parents. Therefore, we match information on parents' parental leave behavior, on further fertility behavior and on labor market outcomes for the time span of over 25 years after the accident. This set of outcomes should be very informative about parents' compensating versus reinforcing investments in their children's human capital. Fifth, we estimate the effect of the early shock on labor market outcomes of the treated cohort; who arrived by now on the labor market. This reveals the impact of the accident (sum of biological effect and parental response) on the productivity of treated children across socio-economic backgrounds. Finally, we carry out an equivalent analysis for siblings of the fall 1986 cohort. Any difference between siblings of treated and control children allow further inference on parents response behavior.

We find that *in utero* exposure to radiation levels (commonly considered harmless) increased the incidence of early fetal death. This culling was more pronounced among low-educated families and also for boys (i. e. we find a significant effect on the sex-ratio). As a consequence, the surviving population exhibits better health outcomes at birth and has a lower likelihood of infant mortality. There is also evidence for prenatal non-radiation effects; highly-educated mothers (in treated and control regions) adjusted their behavior in a way that further distorted the sex-ratio and resulted in a lower weight at birth. We find evidence consistent with negative scarring effects of radiation. In particular we find that mothers of treated children have less post-treatment fertility and reduce their labor supply. Both effects indicate that affected children needed more attention, and parents responded with compensating investment. Especially intriguing is the fact that

³See Bozzoli *et al.* (2009) for a simple model of culling (selection) and scarring.

⁴The identification of the radiation effects relies on random variation in the exposure to radioactive fallout (over time and) between municipalities due to geographic differences in precipitation after the accident. The identification of non-radiation effects rest upon the assumption that behavioral adjustment did not vary with the exposure to radioactive fallout. While the former assumption seems in any case indisputable, the latter assumption can be justified by the fact that actual level of *local* radioactive fallout was verifiably not known at the time of the accident.

the labor supply effect kicks in only after school enrollment. This is consistent with reduced cognitive abilities of treated children and in line with the findings of Almond, Edlund and Palme (2009). This suggests that exposure to radiation entails a positive culling effect on physiological health, and negative scarring effects on cognitive abilities. Results on the labor market outcomes of treated children (and their siblings) are not available yet.

Our results (so far) show that there is a potential large social gradient in the short-run and long-run effects of early-life events that complicate the interpretation of reduced-form estimates. In the case of the Chernobyl accidents we observe three important sources of this social gradient. First, the early shock led to more pronounced culling among loweducated families; which results in unequally selected samples to study later outcomes. Second, (untreated) parents with low and high education processed the emerging uncertain health risk information differently. Third, we observe some differences in the responding investment behavior of parents (of surviving children) across different education groups.

Our findings have important ramifications for the economic literature interested in the effects of early events on the accumulation of human capital. For instance, consider the literature on environmental justice—studying the disproportionately high exposure of individuals with low income to environmental hazards and the resulting impact on their health and economic well-being. Our research design provides the unique opportunity to observe randomly assigned environmental hazards free of any Tiebout sorting on endogenous socio-economic characteristics.⁵ Our findings of substantial treatment effect heterogeneity—where detrimental effects decline along the educational distribution further suggest that in case of conventional environmental hazards (such as air pollution as a byproduct of its production of a marketable good) the average treatment effect on the treated should be higher than the average treatment effect. That means, even results from empirical papers that have a fully credible research design to identify the average treatment effect on the treated (as for instance, based on a difference-in-differences approach; see examples in Greenstone and Gayer (2009)) can not be generalized.

Our results also hold important implications for public policymakers. An informed discussion about the efficiency of nuclear power requires knowledge about the full cost of nuclear and radiation accidents. At least, after the accident in the Fukushima Daiichi Nuclear Power Plant in March 2011, there are serious doubts that even an advanced economy can master nuclear safety. The benefits of nuclear power due to comparable low emissions have to be contrasted not only with the private and social cost involved in the normal operation, but also with the expected total cost of a nuclear accident. Our estimation results provide evidence that accidents in nuclear power planets have large and

⁵This literature typically faces the econometric challenge that exposure to environmental hazards is correlated with a host of confounding factors (Banzhaf and Walsh, 2008) that if unaccounted for lead to an upward bias in the estimates.

long-lasting negative externalities (due to radioactive fallout) even for individuals living about 1,000 miles away; which even translate into reduced economic productivity and income many years after an accident.

The remainder of the paper is organized as follows. Section 2 describes the Chernobyl accident and the resulting radioactive contamination of the western part of the former Soviet Union and Europe. Section 3 presents our identification strategy, the data used, and the econometric specification. Section 4 discusses our findings. Finally, Section 5 concludes the paper.

2 Radioactive contamination of the environment due to the nuclear accident of Chernobyl

On April 26, 1986 at 1:23 A.M. an accident occurred during a systems test at the Chernobyl nuclear power plant in Ukraine (officially the Ukrainian Soviet Socialist Republic) that caused the worst nuclear power plant accident in history. An explosion and fire released large quantities of radioactive contamination into the atmosphere that was not stopped until May 6, 1986.⁶ As a result a plume of highly radioactive fallout spread over an extensive geographical area and drifted in the following days over large parts of the western part of the former Soviet Union and Europe.⁷ The radioactive particles were subsequently removed from the atmosphere solely due to gravitation (*dry deposition*) or by any form of precipitation (*wet deposition*). Given that wet disposition is by far a more efficient deposited on soil and other surfaces (so-called *ground deposition*) was predominantly determined by the presence or absence of precipitation during the passage of the plume (Clark and Smith, 1988).

Radionuclides can enter the human body through inhalation, ingestion, and absorption through the skin. The IAEA (2006, Chapter 5) describes four main pathways by which humans were exposed to the radioactive material released by the accident: (i) external dose from cloud passage, (ii) internal dose from inhalation of the cloud and resuspended material, (iii) external dose from ground deposition, and (iv) internal dose from the consumption of contaminated food and water. The latter two exposure pathways are considered as the most important. That means, humans where exposed to high levels of

⁶This incidence was not immediately announced by the authorities of the Soviet Union, but has been detected after radiation levels set off alarms at a nuclear power plant in Sweden located over one thousand kilometers away from the Chernobyl. The world learned officially about the accident (two days later) on April 28, 1986 through a 20 second announcement by the state television broadcaster in the Soviet Union.

⁷The following link provides a computerized graphic reconstruction of the path of the first 14 days of the radioactive plume, tracking the release of Caesium-137, created by the *Institut de Radioprotection et Sûreté Nucléaire*: http://www.irsn.fr/FR/popup/Pages/tchernobyl_animation_nuage2.aspx.

radiation if they were located in areas with high levels of ground deposition and/or if they consumed large quantities of contaminated food and water. While it is not observable who consumed large quantities of contaminated food and water, the external dose from ground deposition should be highly correlated with the local level of ground deposition at individuals' place of residence. From a researcher's point of view the Chernobyl disaster provides, therefore, an ideal natural experiment to study the effect of the exposure to radioactive ground deposition, since it seems save to assume that the spatial distribution of precipitation during the passage of the plume was random.

The implementation of this research design is facilitated by the wide availability of data on local levels of radioactive ground deposition. In the aftermath of the accident the level of ground deposition of Caesium-137 (henceforth ¹³⁷Cs) and other radionuclides was measured comprehensively on the soil surface in most European countries (European Commission, 1998). In the mapping of the deposition the focus was on ¹³⁷Cs, because it is easy to measure (*ex post*), and because of its radiological significance. It turned out that the three countries (based on current borders) most heavily affected are Belarus, the Russian Federation, and Ukraine. However, Austria, Sweden and Finland also contain some heavily contaminated areas (see, for instance, Figure 3.5. in IAEA (2006)).

3 Research design

In this section we first present the Austrian radiation data we use to distinguish between treated and control units. Then, we present the two most important innovations of our research. First, we highlight the crucial tension between culling and scarring effects that has to be considered in the interpretation of empirical estimates of the effect of the Chernobyl accident, in order *not* to underestimate the true impact. Second, we discuss the potential importance of non-radiation effects of the Chernobyl accident due to behavioral adjustment (of parents). Following that we provide details on the outcomes under consideration, and explain how we translate our research design into a regression framework.

3.1 Radiation data

In Austria radioactive fallout (due to Chernobyl) was measured at 1881 sites, which provides on average of one sample per 45 square kilometers (Bossew *et al.*, 1996, 2001). Radioactive fallout is measured as ground deposition of ¹³⁷Cs (half-life of 30 years) and ¹³⁴Cs (half-life of 2 years) in kilobecquerels (kBq) per square meter (m²).⁸ We focus on average level of ground deposition of ¹³⁷Cs and aggregate measurements to the community-level. This provides us with data for 925 (out of 2, 357) communities, where each data

⁸Dose rate measurements of gamma radiation (in millisievert) immediately after the arrival of the radioactive cloud (based on 336 dose rate meters distributed over the territory of Austria) show a high correlation of dose rates with the deposition measurements of ¹³⁷Cs and ¹³⁴Cs.

point refers to May 1, 1986.⁹

Figure 1 depicts the spatial distribution of contamination, where we distinguish between communities with a ground deposition of 137 Cs below 17 kBq/m², between 17 and 36 kBq/m², above 36 kBq/m², and communities without data.



Figure 1: Average Caesium-137 ground deposition in Austria on May 1, 1986

UNSCEAR (2000) considers regions with 37 kBq/m² of ¹³⁷Cs ground deposition or more as contaminated. In Austria, the average level of contamination was around 23 kBq/m². Communities with the lowest level of contamination recorded only 0.7 kBq/m², while the most contaminated areas had values of about 150 kBq/m². It is this wide range of (within country) variation in radioactive fallout (which is the result of the very local presence or absence of precipitation during the passage of the plume) that makes the Austrian case so particularly well suited for studying the impact of the Chernobyl accident.

In order to define treated and control units, we distinguish between communities (and their residing population) who were exposed to different levels of radioactive fallout. We follow the criteria suggested by UNSCEAR (2000) and define the 175 communities in our sample with 37 kBq/m² of ¹³⁷Cs ground deposition or more as treatment group 1 (*T*1). We specify two further treatment groups with higher levels of contamination. To the 130 communities with 42 kBq/m² of ¹³⁷Cs ground deposition or more we refer as treatment group 2 (*T*2), and to the 93 communities with 47 kBq/m² of ¹³⁷Cs ground deposition or less. Communities with 42 kBq/m² of ¹³⁷Cs ground deposition or less. Communities with medium levels of ¹³⁷Cs ground deposition (i.e. between 17 and 36/41/46 kBq/m²)

 $^{^{9}\}mathrm{Note,}$ the accident happened on April 26, however, the radioactive plume arrived on April 29 in Austria.

are excluded from the analysis. Table 1 summarizes this grouping of communities.

	<u> </u>		8- ° - P~	
Group	Acronym	Average level of 137 Cs ground deposition (in kBq/m ²)	No. of communities	$\substack{\rm Mean}{^{137}\rm Cs}$
Control group	C	less than 17	428	9
$Excluded^a$		between 17 and $36/41/46$	322/367/404	26/27/29
Treatment group 1	T1	37 ore more	175	51
Treatment group 2	T2	42 ore more	130	56
Treatment group 3	T3	47 ore more	93	60

Table 1: Definition of treatment and control groups

a The cutoff-value, number of communities and mean level of 137 Cs depend on the treatment group.

3.2 Estimation strategy

It is conjectured that radiation exposure is especially critical at a prenatal stage. While a human embryo or fetus is protected in the uterus, and the radiation exposure to a fetus should be lower than the dose to its mother, a embryo or fetus is particularly sensitive to ionizing radiation. The empirical evidence on the effects of prenatal exposure on child health is either based on case studies of children born to women who had been treated with high doses of medical radiation while pregnant (De Santis, Di Gianantonio, Straface, Cavaliere, Caruso, Schiavon, Berletti and Clementi, 2005) or on children who have been prenatally close to the hypocenter of the atomic bomb explosions in Hiroshima and Nagasaki (Otake and Schull, 1998; Yamazaki and Schull, 1990).

The possible effects of prenatal radiation exposure include immediate effects (such as fetal death or malformations) or increased risk for cancer later in life. The *Centers for Disease Control and Prevention (CDC)* concludes that gestational age and the radiation dose determine the non-cancer health effects, while the carcinogenic risks are assumed to be constant throughout the pregnancy. Radiation-induced non-cancer health effects are especially detrimental in the first weeks after conception, since an embryo is made up of only a few cells. A damage to one cell (the progenitor of many other cells) may cause the death of the embryo, and the blastocyst will fail to implant in the uterus. Non-cancer health risks are supposed to decrease with gestation length. Beyond about 26 weeks, the fetus is believed to be 'relatively radio-resistent' (i. e. equally sensitive to radiation as a newborn). Given this medical evidence we follow Almond *et al.* (2009) and focus on the impact of prenatal exposure to radiation. However, we do not restrict our sample to children of gestational age 8 to 25 weeks at the time of the accident, but include the entire fall 1986 birth cohort in our analysis.

3.2.1 Radiation & non-radiation effects

There are at least two channels through which the Chernobyl accident may have affected the fall 1986 birth cohort. First, there might be a true causal effect of exposure to radioactive fallout. Second, parents (and potential parents) may have changed their behavior due to the Chernobyl accident. We refer to the former as *radiation effects*, and to the latter as *non-radiation effects*.

A likely behavioral reaction may have been an adaption in fertility behavior. It is plausible that some potential parents decided, due to fear or anxiety after the accident, to postpone family formation by avoiding conception, or even by inducing an abortion. Most likely, already pregnant women where extremely stressed and anxious in the aftermath of the accident, which may had detrimental effects on the embryo or fetus. Experimental evidence from animal studies suggests a link between exogenous in-utero exposure to maternal stress (measured by cortisol levels) and poor offspring outcomes (Kaiser and Sachser, 2005). Aizer *et al.* (2009) show that exposure to elevated levels of the stress hormone cortisol in-utero negatively affects offspring educational attainment and the probability of a severe chronic health condition and verbal IQ at age 7. Currie and Rossin-Slater (2012) finds some evidence for complications of labor and delivery but no effect on birth weight and gestation.

In the case of radiation effects one would expect variation in the estimated effects according to the degree of exposure to radioactive fallout. In the case of non-radiation effects, it is *a priori* not clear whether the degree of behavioral adjustment should vary with exposure. However, given that individuals have not been aware of the local level of ground deposition a uniform response across regions can be expected.

3.2.2 Culling & scarring effects

Radioactive exposure and/or non-radiation effects experienced *in utero* may do more than 'scar' survivors. That means, Chernobyl may have also increased fetal mortality as well as early life mortality rates ('culling'). Survivors exposed to Chernobyl are thus a potentially selected sample, where selection is endogenous to the same adverse events as the scarring effect. This tension between culling and scarring effects has been long recognized in epidemiology. While the scale of the selective mortality is hard to assess we have a clear expectation for its direction. It seems plausible that mortality tends to eliminate those in poor health. In other words, survivors of Chernobyl should generally be positively selected. That means, for the interpretation of the estimated effects we have to keep in mind that negative scarring effects have to be sufficiently strong among the survivors to exceed the positive effect of culling. In order to understand the full impact of Chernobyl we do not only look at the outcomes (of the potentially selected sub-sample) of born children, but we also try to infer the effect on the incidence of conceptions, miscarriages, abortions, stillbirths, and live births.

Clearly, we do not have information on the incidence of sexual intercourse and conceptions. We also do not observe the incidence of miscarriages (medically termed spontaneous abortion). Very early miscarriages (so-called early pregnancy losses) happen in many cases before a woman may know she is pregnant and, therefore, without clinical recognition. Later miscarriages, which occur after the sixth week since the woman's last menstrual period (so-called clinical spontaneous abortion) are unfortunately not universally documented in Austria. Still, miscarriages are a very common phenomena; the incidence of spontaneous abortion is widely believed to be 50 percent of all pregnancies. Unfortunately, we also do not have access to information on induced abortions. As in most countries, Austria begins its comprehensive documentation of fetal mortality with stillbirths. The definition of stillbirths (in particular, the differentiation to miscarriages) varies somewhat across countries. In Austria, a stillbirth is defined as birth of a child of at least 35 centimeter of length, without vital signs. Smaller fetus are categorized as miscarriages, and therefore, not universally documented. Finally, live births are very well documented in the Austrian Birth Register which comprises individual-level data on the parents and the new-born.

In order to infer on the effects of Chernobyl on miscarriages we offer different (complementary) strategies. First, we follow Sanders and Stoecker (2011) and use the sex-ratio of live (and still) births as a metric of fetal death. This methodology is based on an evolutionary theory advocated by Trivers and Willard (1973). The so-called *Trivers-Willard Hypothesis* states that the population sex-ratio responds to parental conditions. In order to maximize the reproductive success of the offspring mothers in good conditions are expected to have more sons, while mothers in poor conditions should have more daughters. This prediction can be rationalized by the assumption that the relationship between health and mating success is less pronounced for women (compared to men). Put differently, since males can in principle have children with multiple women, healthy males could secure several mates, while males in poor health secure none. In contrast, in the case of females, mating with healthy men is also possible for females in poor health.

The precise mechanism how mothers (or their reproductive system) favor either female or male offspring, depending on their condition, is still debated. As prenatal mechanisms, researchers discuss the prevention of the implantation of embryos of certain sex, or the increased likelihood of certain fetal loss. In any case, there exists robust empirical evidence (see, for instance, Almond and Edlund (2007); Catalano, Bruckner, Anderson and Gould (2005); Hansen, Møller and Olsen (1999)) that women in poor health (or under less favorable conditions) are more likely to have female offspring. In our empirical analysis we associate an increased probability of female births with an increase in miscarriages.

Second, we examine all conceptions between August 1, 1984 and July 31, 1987 and distinguish between four different birth cohorts, where some of them are by definition

not affected by Chernobyl via conceptions and/or induced abortions. This is depicted by Figure 2. Birth cohort 0 (BC_0) includes all children who were conceived before August 1, 1985 and most likely born before the Chernobyl accident. In particular, we define birth cohort I (BC_I) as those children who were conceived between August 1, 1985 and January 31, 1986. These children have been *in utero* for more than 3 months at the time of the accident (second & third trimester), which ruled out the option of a induced abortion (except for health-related reasons). That means, BC_I can not be affected by a behavioral response (at the extensive margin) that works through an adapted fertility and/or induced abortion behavior. Children belonging to birth cohort II (BC_{II}) were conceived between February 1, 1986 and April 30, 1986. Since they have been *in utero* for less than 3 months at the time of the accident (first trimester) a behavioral response at the extensive margin via an induced abortion was possible. Children from birth cohort III (BC_{III}) were conceived between May 1, 1986 and July 31, 1987. For these birth cohort a behavioral response at the extensive margin via contraception and/or induced abortion was possible.

Figure 2: Cohorts



Naturally, we do not observe the exact day of conception in our data. Based on the stated gestation length measured in commenced weeks (gl) and the birth day (bd) we compute the conception day (cd) as follows cd = bd - 7 * (gl - 0.5). That means, we assume that a pregnancy with a stated gestation length of 38 weeks has lasted 38.5 weeks or 269.5 days. We exclude conceptions 7 days before and after each cutoff date in order to minimize errors in group assignment.¹⁰ Moreover, we exclude teenage and older mothers and focus on children born to mothers between the age of 20 and 40.

¹⁰Since we classify individuals by conception date (which is measured with error) we can not preclude that, for instance, some children assigned to birth cohort 0 are born after the Chernobyl accident (those with 40 or more weeks of gestation). Excluding conceptions 7 days before and after each cutoff date should minimize this problem.

3.2.3 Outcome variables

The set of outcome variables covered in our estimation analysis allows us to comprehensively evaluate (the social gradient in) the impact of Chernobyl on the outcomes of children born in fall 1986. We examine (all available) health and human capital outcomes that allow us to infer on the effects of the accident at a prenatal stage, at birth, during adolescence, and early adulthood. Furthermore, we examine the postnatal investment behavior of parents (mothers). Therefore, we match data from five main sources of data: (i) The Austrian Birth Register includes the universe of all births in Austria with individual-level information on socioeconomic characteristics and birth outcomes. We use this data to quantify the (socioeconomic group-specific) incidence of live births on a community-level. On an individual-level we examine a set of health indicators including the child's sex, gestation length, birth weight and Apgar scores. (ii) Combining information from the Austrian Death Register allows us to estimate the incidence of stillbirths and infant mortality on the individual-level. (iii) For two Austrian states we have access to the databases of the respective statutory health insurance funds (that cover all private sector employees and their dependents) that allows us to quantify individual health care utilization during adolescence and early adulthood (age 12 to 22 for the cohort *in utero*). (iv) Data from the Austrian Social Security Database allows us to analyze human capital formation and labor market outcomes in early adulthood. In particular, we obtain individual-level information on employment, broad occupation, apprenticeship training, wages and sick leave.

3.2.4 Econometric specification

For all outcomes that are measured on an individual-level our research design translates into the following regression framework, which is performed for each definition of the treatment group T # (T1, T2 and T3):

$$Outcome_{i,c} = \beta_0 + \beta_1 BC_I + \beta_2 BC_{II} + \beta_3 BC_{III} + \beta_4 T \#_{i,c}$$
$$+\beta_5 BC_I \times T \#_{i,c} + \beta_6 BC_{II} \times T \#_{i,c} + \beta_7 BC_{III} \times T \#_{i,c}$$
$$+ \mathbf{X}_{i,c} + \gamma_y + \delta_m + \theta_c + \epsilon_{i,c}$$
(1)

In this equation *i* denotes individual and *c* denotes community. This difference-indifferences (DiD) estimation framework includes binary variables BC_I , BC_{II} and BC_{III} to distinguish between the three birth cohorts as defined above, a binary variable indicating the treatment status of each individual's community $(T \#_{i,c})$, and an interaction term between each birth cohort indicator and the treatment status variable. Further, we control for conception year fixed-effects (γ_y) , conception month fixed-effects (δ_m) and community fixed-effects (θ_c) .¹¹

The parameters β_1 , β_2 , and β_3 give the estimated non-radiation effects for the three different birth cohorts, which differ in scope of potential behavioral adjustment. Therefore, β_1 gives the estimated non-radiation effects (for BC_I) that work through miscarriages. β_2 gives the estimated non-radiation effects (for BC_{II}) due to miscarriages and induced abortions, and β_3 gives the estimated non-radiation effects (for BC_{III}) due to miscarriages, induced abortions and conceptions. That means, if the non-radiation effects on miscarriages would be equal for BC_I and BC_{II} , then the difference between β_2 and β_1 would give us the abortion effect. Equivalently, if non-radiation effects on miscarriages and induced abortions would be equal for BC_{II} and BC_{III} , then the difference between β_3 and β_2 would give us the conception effect. Given that non-radiation effects are constant across treated and control communities, the parameters β_5 , β_6 , and β_7 provide the estimated radiation effects for the different birth cohorts.

For the outcome that is measured on a monthly community-level (i. e. live births) we drop the index i from the regression framework presented in equation (1).

4 Preliminary estimation results

Please note this section is not fully written up yet, and the estimations for some outcomes (see comments below) are still under progress.

4.1 Fertility and perinatal outcomes

4.1.1 Live births & fetal death

Table 3 summarizes the estimated effects on live births from monthly community-level regressions. We exploit the available information on mother's education to split the sample into children born to low educated mothers (with compulsory schooling or less) and children born to highly educated mothers (with an educational attainment higher than compulsory schooling). We do not find any statistically significant non-radiation effects. However, for BC_{II} (which were in the first trimester post conception at the time of the accident) we find a statistically significant negative radiation effect for low educated mothers. The effect amounts to 8.6 to 11.1 percent fewer live births in exposed communities and is significant for all three definitions of exposure to radiation ($T_1 - T_3$). If we further split the sample by sex of the child we find a negative (although less significant) effect for both sexes.¹² The radiation effect is smaller and less significant for children in the second

¹¹Note that the treatment indicator $T\#_{i,c}$ is dropped because of perfect collinearity with the community fixed-effects θ_c .

 $^{^{12}\}mathrm{These}$ results are available upon request.

or third trimester post conception (BC_I) which is consistent with the existing evidence on the heterogenous impact of radioactive exposure over the gestation period (see above).

A reduced incidence of live births suggests strong culling. In other words, this would imply that either some children are stillbirths or that embryos and/or fetus die at an earlier stage (or both), and the surviving population is positively selected. Table 4 summarizes the estimated effects on stillbirths from individual-level regressions. We do not find any statistically significant effects, neither radiation nor non-radiation effects. This last result suggests that radioactive exposure should lead to a higher incidence of fetal death. While we can not directly observe early fetal death, we can use the sex-ratio as a proxy variable. As the results summarized in Table 5 show we indeed find a statistically significant negative effect on the sex-ratio in BC_{II} . Within the group of low educated mothers, exposure to radiation in the first trimester post conception leads to 4.2 to 8.8 percentage points fewer male births. The size and significance of the effect increases with the level of radiation. Following the literature and interpreting this as evidence for miscarriages, this finding is consistent with the negative effect on live births (and the zero effect on stillbirths).

These results further suggest that the relevant prenatal mechanism that distorts the sex-ratio is male fetal loss and not the prevention of the implantation of embryos of a certain sex. If male fetus (embryos) die at a very early stage – i.e. the mother's body already prevents the implantation of male embryos – then the male embryos can get replaced (without much time lag) by a female embryo at the next conception, and the sex-ratio may change with leaving the incidence of live births constant. However, if the sex-ratio gets distorted at a later stage via male fetal loss, the incidence of live births should decrease, as we observe.

In sum this set of results provides robust evidence that prenatal radiation exposure (of about 137 Cs ground deposition of 37 kBq/m² or higher) increases the likelihood of prenatal death, and supports existing medical literature. Our analysis reveals that male embryos and fetus are more vulnerable to radiation compared to female ones, and prenatal radiation exposure distorts the sex-ratio substantially. Notably, the detrimental effects of prenatal radiation exposure on prenatal mortality seem to include a social gradient. We find very robust evidence that the decrease in the number of live births and the decreased likelihood of male live births (our proxy for prenatal death) prevails only among low educated mothers. The finding that radiation seems to harm (in this dimension) only mothers from a lower socio-economic background is consistent with research on differences in the reaction of low and highly educated individuals to emerging health risk information. For instance, Aizer and Stroud (2010) show that highly educated women immediately reduced smoking in response to the 1964 Surgeon General Report on Smoking and Health while the low educated did not, and Anderberg, Chevalier and Wadsworth (2011) find evidence for a differential response of low and highly educated parents to the measles, mumps and rubella (MMR) controversy in the UK.

4.1.2 Health at birth

In order to evaluate the impact of Chernobyl on the health at birth we have individuallevel data on gestation length, birth weight, and Apgar scores after one, five and ten minutes available.¹³ For the interpretation of this set of health outcomes we have to keep the tension between culling and scarring effects in mind. Since we only observe the survivors—the most likely positively selected group of healthier newborns—our estimated radiation (and non-radiation) effects reflect the sum of a positive culling and a negative scarring effect. Positive estimates would reflect a very strong culling which overcompensates possible scarring effects. Negative estimates would indicate the dominance of scarring. Statistically insignificant effects could either mean that culling and scarring cancel each other out, or the lack of any causal effect of Chernobyl.

The estimation results on the effects on gestation length (see Tables 6 and 7) and birth weight (see Tables 9 and 8) provide evidence for a positive culling through radiation effects (that overcompensates any scarring effects, if existent). Most consistent with the results on live births and fetal death we find that children born to low educated mothers are about 3 percentage points less likely to be preterm births, i. e. born before a gestational age of 37 weeks. The effect is less strong for total gestation length in days and amounts to 1.5 additional days of gestation on average. In the case of birth weight, culling is less evident. Children of low educated mothers do not have a significantly higher birth weight or a lower probability of having low birth weight (defined as weighing less than 2500 grams).

Somewhat surprisingly, for some outcomes a similar pattern is also observed among children born to highly educated mothers. For all birth cohorts we see a radiation effect of about an additional day of gestation; the exact point estimates tend to increase with the level of radioactive exposure. Again, this results suggests that culling effects dominate. However, we find no evidence for a decreased probability of preterm birth. In the case of birth weight the positive effects are concentrated at BC_{II} with estimated effects of 4-6 decagrams—depending on the level of radioactive exposure.

For children born to highly educated mothers we also find some evidence for nonradiation effects. Our estimation results suggest that behavioral adjustments resulted in a lower sex-ratio and a lower birth weight in BC_I .¹⁴ We find that children form this birth cohort exhibit (uniformly across regions with different levels of contamination) a reduced birth weight of minus three decagrams. This might be the result of a changed diet of pregnant women and is consistent with evidence for a higher responsiveness of birth weight to nutritional changes in the third trimester of pregnancy (e. g. Painter *et al.*, 2005).

¹³The Apgar score is based on five criteria (appearance, pulse, grimace, activity and respiration) and ranges from zero to ten.

¹⁴The effect on birth weight in BC_I is not due to the lower likelihood of having a male child and is still significant if we condition on the sex of the child.

For the Apgar scores (in the paper we present detailed output for the score after ten minutes, see Table 10) we find little evidence for radiation or non-radiation effects for neither socioeconomic group. For BC_I we find some evidence for scarring of children of highly educated mothers in communities with comparably higher levels of radioactive exposure.

In sum these results indicate that the surviving population of children seem to be positively selected with respect to health at birth.

4.1.3 Infant mortality

Please note the estimation results on the effect on infant mortality are available in Tables 11 to 14. This subsection, however, still needs to be written up.

4.2 Parental investments

Tables?? to 19 provide evidence for postnatal responsive investment behavior of the surviving kids' mothers. More to follow.

4.3 Child outcomes later in life

This section is still in progress.

5 Conclusions

This section is still in progress.

References

- AIZER, A. and STROUD, L. (2010). Education, Knowledge and the Evolution of Disparities in Health. Working paper 15840, National Bureau of Economic Research, Cambridge, MA.
- —, and BUKA, S. (2009). *Maternal Stress and Child Well-Being: Evidence from Siblings*. Working paper, Brown University Department of Economics.
- ALMOND, D. and CURRIE, J. (2011). Human Capital Development before Age Five. In O. Ashenfelter and D. Card (eds.), *Handbook of Labor Economics*, Elsevier, vol. 4, pp. 315–1486.
- and EDLUND, L. (2007). Trivers-Willard at Birth and One Year: Evidence from US Natality Data 1983-2001. Proceedings of the Royal Society B: Biological Sciences, 274 (3), 2491–2496.
- —, and PALME, M. (2009). Chernobyl's Subclinical Legacy: Prenatal Exposure to Radioactive Fallout and School Outcomes in Sweden. The Quarterly Journal of Economics, **124** (4), 1729–1772.
- ANDERBERG, D., CHEVALIER, A. and WADSWORTH, J. (2011). Anatomy of a health scare: Education, income and the MMR controversy in the UK. *Journal of Health Economics*, **30** (3), 515–530.
- AUVINEN, A., VAHTERISTO, M., ARVELA, H., SUOMELA, M., RAHOLA, T., HAKAMA, M. and RYTÖMAA, T. (2001). Chernobyl Fallout and Outcome of Pregnancy in Finland. *Environmental Health Perspectives*, **109** (2), 179–185.
- BANZHAF, H. and WALSH, R. P. (2008). Do People Vote with Their Feet? An Empirical Test of Tiebout. *American Economic Review*, **98** (843–863), 3.
- BERTOLLINI, R., LALLO, D. D., MASTROIACOVO, P. and PERUCCI, C. A. (1990). Reduction of Births in Italy after the Chernobyl Accident. Scandinavian Journal of Work, Environment & Health, 16 (2), 96–101.
- BLETTNER, M. (2000). European Stillbirth Proportion and Chernobyl. International Journal of Epidemiology, **29** (3), 596–599.
- BOSSEW, P., DITTO, M., FALKNER, T., HENRICH, E., KIENZL, K. and RAPPELS-BERGER, U. (1996). *Cäsiumbelastung der Böden Österreichs*. Tech. rep., Umweltbundesamt.
- —, —, —, —, and (2001). Contamination of Austrian Soil with Caesium-137. Journal of Environmental Radioactivity, **55** (2), 187–194.

- BOZZOLI, C., DEATON, A. and QUINTANA-DOMEQUE, C. (2009). Adult Height and Childhood Disease. *Demography*, **46** (4), 647–669.
- CATALANO, R., BRUCKNER, T., ANDERSON, E. and GOULD, J. B. (2005). Fetal Death Sex Ratios: A Test of the Economic Stress Hypothesis. *International Journal of Epidemiology*, 34, 944–948.
- CLARK, M. J. and SMITH, F. B. (1988). Wet and Dry Deposition of Chernobyl Releases. *Nature*, **332**, 245–249.
- CONTI, G., HECKMAN, J. J., YI, J. and ZHANG, J. (2011). *Early Health Shocks, Parental Responses, and Child Outcomes.* Unpublished manuscript, University of Chicago.
- CURRIE, J. and ROSSIN-SLATER, M. (2012). Weathering the Storm: Hurricanes and Birth Outcomes. Working paper 18070, National Bureau of Economic Research, Cambridge, MA.
- DE SANTIS, M., DI GIANANTONIO, E., STRAFACE, G., CAVALIERE, A. F., CARUSO, A., SCHIAVON, F., BERLETTI, R. and CLEMENTI, M. (2005). Ionizing Radiations in Pregnancy and Teratogenesis: A Review of Literature. *Reproductive Toxicology*, 20, 323–329.
- ERICSON, A. and KALLEN, B. (1994). Pregnancy Outcome in Sweden after the Chernobyl Accident. *Envirionmental Research*, **67**, 149–159.
- EUROPEAN COMMISSION (1998). Atlas of Caesium Deposition on Europe After the Chernobyl Accident. Atlas, Office for Official Publications of the European Communities, Luxembourg.
- GREENSTONE, M. and GAYER, T. (2009). Quasi-Experimental and Experimental Approaches to Environmental Economics. Journal of Environmental Economics and Management, 57 (21–44), 1.
- GROSCHE, B., IRL, C., SCHOETZAU, A. and VAN SANTEN, E. (1997). Perinatal Mortality in Bavaria, Germany, After the Chernobyl Reactor Accident. *Radiation and Environmental Biophysics*, **36**, 129–136.
- HAEUSLER, M. C., BERHOLD, A., SCHOELL, W., HOFER, P. and SCHAFFER, M. (1992). The Influence of the Post-Chernobyl Fallout on Birth Defects and Abortion Rates in Austria. American Journal of Obstetrics & Gynecology, 167 (4), 1025–1031.
- HANSEN, D., MØLLER, H. and OLSEN, J. (1999). Severe Periconceptional Life Events and the Sex Ratio in Offspring: Follow Up Study Based on Five National Registers. *British Medical Journal*, **319**, 548–549.

- HARJULEHTO, T., ARO, T., RITA, H., RYTÖMAA, T. and SAXÉN, L. (1989). The Accident at Chernobyl and Outcome of Pregnancy in Finland. *British Medical Journal*, 298, 995–997.
- HOLT, E. (2010). Debate Over Health Effects of Chernobyl Re-ignited. *The Lancet*, **375** (9724), 1424–1425.
- IAEA (2006). Environmental Consequences of the Chernobyl Accident and Their Remediation: Twenty Years of Experience. Report of the Chernobyl Forum Expert Group "Environment", International Atomic Energy Agency, Vienna.
- IRGENS, L. M., LIE, R., ULSTEIN, M., JENSEN, T. S., SKJÆRVEN, R., SIVERTSEN, F., REITAN, J. B., STRAND, F., STRAND, T. and SKJELDESTAD, F. E. (1991). Pregnancy Outcome in Norway after Chernobyl. *Biomedicine & Pharmacotherapy*, 45 (6), 233–241.
- KAISER, S. and SACHSER, N. (2005). The Effects of Prenatal Social Stress on Behaviour: Mechanisms and Function. *Neuroscience and Biobehavioral Reviews*, **29** (2), 283–294.
- KÖRBLEIN, A. and KÜCHENHOFF, H. (1997). Perinatal mortality in germany following the chernobyl accident. *Radiation and Environmental Biophysics*, **36**, 3–7.
- LITTLE, J. (1993). The Chernobyl Accident, Congenital Anomalies and Other Reproductive Outcomes. *Paediatric and Perinatal Epidemiology*, 7, 121–151.
- LÜNING, G., SCHMIDT, M., SCHEER, J. and ZIGGEL, H. (1989). Early Infant Mortality in West Germany Before and After Chernobyl. *The Lancet*, **334** (8671), 1081–1083.
- OTAKE, M. and SCHULL, W. (1998). Review: Radiation-related Brain Damage and Growth Retardation Among the Prenatally Exposed Atomic Bomb Survivors. *Interna*tional Journal of Radiation Biology, 74 (2), 159–171.
- PAINTER, R. C., ROSEBOOMA, T. J. and BLEKER, O. P. (2005). Prenatal Exposure to the Dutch Famine and Disease in Later Life: An Overview. *Reproductive Toxicology*, 20, 345–352.
- SANDERS, N. J. and STOECKER, C. (2011). Where Have All the Young Men Gone? Using Gender Ratios to Neasure the Effect of Pollution on Fetal Death Rates. Unpublished manuscript.
- SCHERB, H., WEIGELT, E. and BRÜSKE-HOHLFELD, I. (1999). European Stillbirth Proportions Before and After the Chernobyl Accident. International Journal of Epidemiology, 28 (5), 932–940.

- —, and BRÜSKE-HOHLFELD, I. (2000). Regression Analysis of Time Trends in Perinatal Mortality in Germany, 1980-1993. Environmental Health Perspectives, 108 (2), 159–165.
- TRICHOPOULOS, D., ZAVITSANOS, X., KOUTIS, C., DROGARI, P., PROUKAKIS, C. and PETRIDOU, E. (1987). The Victims of Chernobyl in Greece: Induced Abortions After the Accident. *British Medical Journal*, **295**, 1100.
- TRIVERS, R. L. and WILLARD, D. E. (1973). Natural Selection of Parental Ability to Vary the Sex Ratio of Offspring. *Science*, **179** (4068), 90–92.
- UNSCEAR (2000). Report to the General Assembly: Sources and Effects of Ionizing Radiation, Vol. II, Annex J. Unscear 2000 report, United Nations Scientific Committee on the Effects of Atomic Radiation, New York.
- WHO (2006). Health effects of the Chernobyl accident and special health care programmes. Report of the UN Chernobyl Forum Expert Group "Health", World Health Organization, Geneva.
- YAMAZAKI, J. N. and SCHULL, W. J. (1990). Perinatal Loss and Neurological Abnormalities Among Children of the Atomic Bomb. Nagasaki and Hiroshima Revisited, 1949 to 1989. Journal of the American Medical Association, 264 (5), 605–609.

6 Tables and Figures

Outcome	Level of available data	Measurements	Data source
Conceptions	not observable	T	ſ
Live births	community	absolute number per 1,000 female inhabitants aged 15-39 in 1981 (in the respective education group)	Austrian Birth Register
Stillbirth	individual	binary variable	Austrian Birth Register
Spontaneous abortions	not observable	proxied by sex-ratio	Austrian Birth Register
Induced abortions	not observable	1	1
Health at birth	individual	gestation length, weight, Apgar scores, etc.	Austrian Birth Register
Infant mortality	individual	binary variables that indicate whether child is still alive after 24 hours/7 days/1 month/1 year	Austrian Birth & Death Register
Health later in life	individual	health care utilization	Health Insurance Funds
Labor market (c, m)	individual	employment, occupation, wages	Austrian Social Security Database
Pre-birth maternity leave	individual	length in days	Austrian Social Security Database
Post-birth maternity leave	individual	length in days	Austrian Social Security Database
Parental leave	individual	length in days	Austrian Social Security Database
Post-treatment fertility	individual	number of children	Austrian Birth Register/ Austrian Social Security Database

 Table 2: Potential outcomes

	Low e	educated mc	others	Highly	educated m	others
	$37 \ \mathrm{kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	47 kBq/m^2	$37 \ \rm kBq/m^2$	$42 \ \rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$
BCI	0.034	0.034	0.036	-0.101	-0.084	-0.109
	(0.033)	(0.034)	(0.035)	(0.076)	(0.079)	(0.081)
BC II	0.035	0.035	0.034	0.058	0.080	0.030
	(0.057)	(0.060)	(0.062)	(0.129)	(0.134)	(0.137)
BC III	0.066	0.062	0.064	-0.158	-0.128	-0.187
	(0.066)	(0.070)	(0.073)	(0.151)	(0.158)	(0.161)
T # x BCI	-0.044	-0.082*	-0.092	0.036	0.083	-0.003
	(0.040)	(0.046)	(0.058)	(0.086)	(0.101)	(0.125)
T # x BC II	-0.086**	-0.111^{**}	-0.106^{*}	-0.060	0.053	-0.025
	(0.042)	(0.049)	(0.057)	(0.125)	(0.152)	(0.177)
T # x BC III	0.021	0.006	0.023	0.130^{**}	0.144^{**}	0.136^{*}
	(0.029)	(0.032)	(0.038)	(0.066)	(0.068)	(0.076)
Abortion effect	0.001	0.000	-0.002	0.159	0.164	0.139
P-value	0.985	0.995	0.966	0.146	0.146	0.232
Conception effect	0.031	0.028	0.030	-0.216	-0.208	-0.217
P-value	0.359	0.433	0.411	0.014	0.020	0.018
Obs.	21636	20016	18684	21600	19980	18648
Mean (absolute)	3.031	3.066	3.076	8.918	8.902	8.873
This table summarizes	estimation resu	lts based on mo	nthly communit	y level data com	puted from the	Austrian Birth
Register in the period	from 08/1984 th	nrough 07/1987.	The dependen	t variable is equ	al to the number	er of live births
Per 1,000 tentate muta Method of estimation	is least squares	The control of	onort was conc	suucauon group) ceived between () uiviueu by tiit 08/1984 and 07	/1985 and was
born before the Chern	lobyl accident.	BC_I was concei	ived between 08	3/1985 and 01/1	986 and was be	etween 4 and 9
months post conception	on at the time c	of the accident.	BC_{II} was con	ceived between	02/1986 and 04	/1986 and was
between 0 and 3 mont	hs post conception	on at the time o	f the accident.	BC_{III} was conc	eived after Cher	nobyl (between
00/1900 and 01/1901). concention-month five	Lacn DiU specii d-effects Bohiist	t standard error	ior the number (s in narenthese	or innabitants, cc * * ** and ***	ommunty, conce indicate statisti	puon-year, and cal significance
at the 10-percent level	, 5-percent level,	and 1-percent.	warming in a			

Table 3: The effect of the accident on live births (elasticities)

	Low 6	educated mc	others	Highly	educated m	others
	$37 \ \mathrm{kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$
BC I	0.001	0.001	0.002	0.002	0.002	0.002
	(0.003)	(0.003)	(0.003)	(0.001)	(0.001)	(0.001)
BC II	-0.006	-0.004	-0.004	-0.001	-0.001	-0.001
	(0.006)	(0.005)	(0.006)	(0.003)	(0.003)	(0.003)
BC III	-0.006	-0.003	-0.003	0.000	-0.000	-0.000
	(0.006)	(0.006)	(0.006)	(0.003)	(0.003)	(0.003)
T # x BCI	0.000	0.001	0.001	0.001	0.000	0.000
	(0.003)	(0.004)	(0.004)	(0.002)	(0.002)	(0.002)
T#xBCII	0.004	0.009	0.009	-0.001	0.000	-0.002
	(0.005)	(0.007)	(0.008)	(0.002)	(0.002)	(0.002)
T#xBCIII	0.000	0.003	0.004	-0.000	0.000	0.000
	(0.003)	(0.003)	(0.003)	(0.001)	(0.001)	(0.002)
Abortion effect	-0.008	-0.005	-0.005	-0.002	-0.003	-0.003
P-value	0.134	0.295	0.280	0.356	0.284	0.240
Conception effect	0.001	0.000	0.001	0.001	0.001	0.001
P-value	0.837	0.887	0.842	0.627	0.617	0.524
Obs.	25986	24163	22917	71695	65610	61914
Mean	0.005	0.005	0.005	0.004	0.004	0.004
This table summarizes	s estimation resul	ts based on indiv	vidual-level data	h from the $Austr$	ian Birth Regist	er in the period
from $08/1984$ through	07/1987. The d	ependent variabl	le is equal to on	e if the child is a	a stillbirth and	zero if the child
is a live birth. Method	d of estimation is	s a LMP. The cc	introl cohort we	s conceived betw	ween 08/1984 ai	nd 07/1985 and
was born before the C	tion of the time	1t. BC_I was color ⁴	DCDC	1 08/1985 and 0.	1/1986 and was	between 4 and
between 0 and 3 mont	ths nost concention	on at the time o	f the accident.	BC_{III} was conce	oz/1300 autu 05 eived after Cher	e/ 1900 aud was nohvl (hetween
05/1986 and $07/1987$)	. Each DiD spec	ification controls	s for community	, conception-yea	r, and conceptic	on-month fixed-
effects. Robust stands	ard errors in pare	entheses. *, ** a	and *** indicate	e statistical signi	ficance at the 1	0-percent level,
5-percent level, and 1-	percent.					

Table 4: The effect of the accident on stillbirths (elasticities)

	Low e	educated mc	others	Highly	educated m	others
	$37~{\rm kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	$47 \ \rm kBq/m^2$	$37 \ \rm kBq/m^2$	$42~{\rm kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$
BCI	0.005	0.006	0.016	-0.020*	-0.024**	-0.022*
	(0.018)	(0.018)	(0.018)	(0.012)	(0.012)	(0.012)
BC II	0.040	0.043	0.056	0.022	0.013	0.016
	(0.034)	(0.035)	(0.036)	(0.023)	(0.023)	(0.023)
BC III	0.017	0.020	0.041	-0.017	-0.025	-0.021
	(0.033)	(0.034)	(0.035)	(0.024)	(0.024)	(0.024)
T # x BCI	0.006	0.009	0.007	-0.007	-0.005	0.001
	(0.022)	(0.026)	(0.033)	(0.013)	(0.016)	(0.021)
T # x BC II	-0.042	-0.074^{*}	-0.088**	-0.014	-0.005	-0.015
	(0.036)	(0.039)	(0.040)	(0.019)	(0.022)	(0.026)
T # x BC III	0.008	0.008	0.012	-0.007	0.001	0.000
	(0.018)	(0.021)	(0.024)	(0.010)	(0.011)	(0.013)
Abortion effect	0.035	0.038	0.040	0.042	0.037	0.038
P-value	0.243	0.224	0.202	0.019	0.044	0.039
Conception effect	-0.023	-0.023	-0.015	-0.039	-0.038	-0.037
P-value	0.252	0.258	0.463	0.000	0.000	0.001
Obs.	25861	24050	22810	71436	65374	61691
Mean	0.511	0.511	0.511	0.514	0.515	0.514
This table summarized	s estimation res	ults based on i	ndividual-level	data from the .	Austrian Birth	Register in the
Deriod from U8/1984 Method of estimation i	through U7/1987 is a LPM The C	(. The dep. va mtrol cohort wa	ar. is equal to s conceived bet	one if the child ween O8/1984 an	1 is male, and d 07/1985 and	zero otherwise. was born before
the Chernobyl acciden	t. BC_I was con	iceived between	08/1985 and 0	1/1986 and was	between 4 and	9 months post
conception at the time	e of the accident	. BC_{II} was cor	nceived between	$02/1986$ and 0^2	4/1986 and was	between 0 and
3 months post concept	tion at the time	of the accident.	BC_{III} was co	nceived after Ch	ernobyl (betwe	= 05/1986 and
07/1987). Each DiD spectrone in none	becification contraction contracti	ols for communi	ty, conception-y	ear and concepti frames of the 10	ion-month fixed ७४ ६७४ वस्ते 1९४	effects. Robust
Stalluaru errois III part	enureses. , a	III IIINICANA	SUBUISUICAL SIGUL	ICALICE AL LILE TU	70, 9.70, anu 1.70	level.

\mathbf{birth}
male
of a
probability
on the
accident a
of the
effect
The
Table 5:

weeks
37
V
(gestation
\mathbf{birth}
premature
of
pq
likelihoo
\mathbf{the}
on
accident
\mathbf{the}
JC
it c
effec
The
6:
Table

	Low e	educated mo	others	Highly	educated m	others
	$37 \ \mathrm{kBq/m^2}$	$42 \ \mathrm{kBq}/\mathrm{m}^2$	$47 \ \mathrm{kBq/m^2}$	$37~{\rm kBq/m^2}$	$42 \ \mathrm{kBq}/\mathrm{m}^2$	$47 \ \mathrm{kBq/m^2}$
BC I	0.003	0.005	0.005	-0.004	-0.004	-0.003
	(0.010)	(0.010)	(0.010)	(0.006)	(0.007)	(0.007)
BC II	0.003	0.001	0.004	-0.008	-0.006	-0.003
	(0.016)	(0.016)	(0.017)	(0.008)	(0.009)	(0.009)
BC III	0.005	0.006	0.009	-0.015	-0.014	-0.010
	(0.020)	(0.021)	(0.022)	(0.011)	(0.012)	(0.012)
T # x BCI	0.007	-0.001	-0.000	-0.005	-0.006	-0.00
	(0.011)	(0.012)	(0.014)	(0.006)	(0.007)	(0.008)
T # x BC II	-0.026^{**}	-0.025^{**}	-0.030^{**}	-0.008	-0.009	-0.011
	(0.011)	(0.013)	(0.014)	(0.008)	(0.00)	(0.010)
T# x BC III	0.009	0.007	0.004	-0.000	-0.002	0.003
	(0.007)	(0.009)	(0.010)	(0.004)	(0.005)	(0.006)
Abortion effect	0.000	-0.003	-0.001	-0.004	-0.001	0.001
P-value	0.999	0.827	0.948	0.570	0.848	0.944
Conception effect	0.003	0.005	0.005	-0.007	-0.008	-0.007
P-value	0.829	0.682	0.691	0.337	0.245	0.314
Obs.	25861	24050	22810	71436	65374	61691
Mean	0.057	0.056	0.057	0.047	0.046	0.046
This table summarizes	estimation resul	ts based on indi	vidual-level data	i from the $Austr$	ian Birth Regist	er in the period
from $08/1984$ through	07/1987. The e	dep. var. is eq	ual to one if the	e gestation peric	pd is below 37 v	veeks, and zero
otherwise. Method of ϵ	estimation is a L	PM. The contro	ol cohort was co	nceived between	108/1984 and 0 ⁷	7/1985 and was
born before the Chern	nobyl accident.]	BC I was conce	ived between 08	3/1985 and $01/5$	1986 and was b	etween 4 and 9
months post conceptio	on at the time o	f the accident.	BC II was con	ceived between	02/1986 and 04	t/1986 and was
between 0 and 3 month	hs post conception	on at the time o	of the accident.	3C III was conc	eived after Cher	nobyl (between
05/1986 and $07/1987$).	Each DiD speci	fication controls	s for community,	year and month	n fixed-effects. B	tobust standard
errors in parentheses.	*, ** and *** inc	dicate statistica	l significance at	the $10\%, 5\%$, a	nd 1% level.	

	Low e	ducated mc	others	Highly	educated m	others
	$37 \ \mathrm{kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	$47 \ \rm kBq/m^2$	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$
BCI	-0.529	-0.540	-0.601	-0.357	-0.321	-0.310
	(0.537)	(0.556)	(0.570)	(0.268)	(0.277)	(0.280)
BC II	-0.233	-0.113	-0.267	-0.584	-0.709	-0.726
	(0.844)	(0.889)	(0.922)	(0.457)	(0.475)	(0.481)
BC III	-0.450	-0.356	-0.505	-0.080	-0.130	-0.139
	(1.092)	(1.147)	(1.193)	(0.521)	(0.556)	(0.566)
T # x BCI	-0.301	-0.189	-0.038	0.943^{***}	1.019^{**}	1.029^{**}
	(0.636)	(0.727)	(0.853)	(0.333)	(0.416)	(0.483)
T # x BC II	1.664^{**}	1.402	1.551	0.857^{**}	0.907^{**}	0.958^{*}
	(0.766)	(0.856)	(0.943)	(0.396)	(0.460)	(0.507)
T # x BC III	0.067	0.136	0.316	0.683^{***}	0.731^{***}	0.542^{*}
	(0.493)	(0.489)	(0.568)	(0.234)	(0.282)	(0.317)
Abortion effect	0.297	0.427	0.334	-0.227	-0.388	-0.416
P-value	0.695	0.589	0.683	0.573	0.346	0.315
Conception effect	-0.217	-0.243	-0.237	0.503	0.578	0.587
P-value	0.716	0.691	0.706	0.095	0.062	0.063
Obs.	25861	24050	22810	71436	65374	61691
Mean	273.164	273.218	273.215	273.815	273.856	273.889
This table summarized	s estimation res	ults based on i	ndividual-level	lata from the .	Austrian Birth	Register in the
period from 08/1984 t	hrough 07/1987.	The dep. var.	is equal to the	e gestation lengt	th in days, and	zero otherwise.
the Chernobyl acciden	t. BC I was con	nceived between	08/1985 and 0	veeн 00/1904 ан 1/1986 and was	between 4 and	9 months post
conception at the time	e of the accident	. BC II was con	nceived between	$02/1986$ and 0^{-1}	4/1986 and was	between 0 and
3 months post concept	tion at the time	of the accident.	BC III was co	nceived after Ch	iernobyl (betwee	05/1986 and
07/1987). Each DiD s	specification cont *** :d: :to cto	trols for commu	nity, year and 1	nonth fixed-effe	cts. Robust sta	ndard errors in
parentneses. ', ' and	Indicate sta	uisticai signinca	nce at the 10%,	070, and 170 lev	eı.	

days
(in
length
l
gestatior
\mathbf{the}
on
accident
f the
0
effect
The
ų.
Table

	Low e	educated mc	others	Highly	educated m	others
	$37 \ \rm kBq/m^2$	$42 \ \rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42 \ \rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$
BCI	-1.246	-0.282	-0.719	-3.131***	-3.464***	-3.350***
	(2.425)	(2.508)	(2.595)	(1.102)	(1.136)	(1.142)
BC II	5.721	7.962^{*}	7.526^{*}	-1.016	-2.239	-2.451
	(3.954)	(4.085)	(4.224)	(1.843)	(1.904)	(1.953)
BC III	0.952	3.400	2.836	-2.461	-3.740^{*}	-3.729*
	(4.708)	(4.943)	(5.135)	(2.053)	(2.137)	(2.174)
T # x BCI	-3.678	-3.818	-3.207	1.892	2.395	2.045
	(2.514)	(3.265)	(3.468)	(1.291)	(1.616)	(1.691)
T#xBCII	0.738	0.219	3.640	3.859^{**}	6.071^{***}	5.828^{**}
	(3.003)	(3.529)	(3.788)	(1.922)	(2.332)	(2.575)
T#xBCIII	0.198	0.778	2.326	0.971	0.553	-0.214
	(1.851)	(1.991)	(2.344)	(1.126)	(1.429)	(1.388)
Abortion effect	6.967	8.243	8.245	2.116	1.225	0.899
P-value	0.031	0.014	0.016	0.207	0.482	0.619
Conception effect	-4.769	-4.561	-4.690	-1.445	-1.501	-1.278
P-value	0.041	0.060	0.059	0.271	0.268	0.353
Obs.	25861	24050	22810	71436	65374	61691
Mean	325.355	325.339	325.142	327.580	327.505	327.453
This table summarized	s estimation res	ults based on i	ndividual-level	data from the .	Austrian Birth	Register in the
period from 08/1984 t Mothod of octimation i	hrough 07/1987.	The dep. var.	is equal to the	birth weight in 2007/108/ an	decagram, and	zero otherwise.
the Chernobyl acciden	t. BC I was cor	iceived between	08/1985 and 0	1/1986 and was	between 4 and	9 months post
conception at the time	e of the accident	. BC II was con	nceived between	$02/1986$ and 0^{-1}	4/1986 and was	between 0 and
3 months post concept	tion at the time	of the accident.	BC III was co	nceived after Ch	iernobyl (betwee	05/1986 and
07/1987). Each DiD : narentheses * ** and	specification cont *** indicate sta	trols for commu tistical significa	nity, year and 1 nce at the 10%	nonth fixed-effe 5% and 1% lev	cts. Robust sta el	ndard errors in
har anna da anna anna	nne onnomit	INTERNET OF PRETINCE	in an and in the total	0/0, atta 1/0 101		

Table 8: The effect of the accident on the birth weight (in decagram)

	Low e	ducated mo	thers	Highly	educated m	others
	$37 \ \mathrm{kBq/m^2}$	$42~\rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$	$37~{\rm kBq/m^2}$	$42 \ \rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$
BCI	-0.005	-0.004	-0.001	0.007	0.008	0.007
	(0.010)	(0.010)	(0.011)	(0.005)	(0.006)	(0.006)
BC II	-0.020	-0.027	-0.019	-0.001	0.003	0.003
	(0.018)	(0.019)	(0.019)	(0.008)	(0.00)	(0.009)
BC III	-0.008	-0.011	-0.002	-0.001	0.003	0.002
	(0.024)	(0.026)	(0.027)	(0.010)	(0.010)	(0.010)
T # x BCI	0.019	0.014	0.015	-0.011^{*}	-0.008	-0.003
	(0.012)	(0.015)	(0.017)	(0.007)	(0.008)	(0.00)
T#xBCII	-0.012	-0.009	-0.020	-0.026^{***}	-0.032***	-0.028***
	(0.013)	(0.015)	(0.016)	(0.007)	(0.008)	(0.009)
T#xBCIII	0.007	0.008	0.013	-0.004	-0.002	0.005
	(0.008)	(0.010)	(0.012)	(0.005)	(0.007)	(0.006)
Abortion effect	-0.016	-0.022	-0.018	-0.008	-0.004	-0.004
P-value	0.275	0.135	0.244	0.313	0.591	0.644
Conception effect	0.013	0.016	0.017	-0.000	-0.001	-0.001
P-value	0.302	0.200	0.182	0.963	0.913	0.839
Obs.	25861	24050	22810	71436	65374	61691
Mean	0.067	0.067	0.067	0.054	0.054	0.055
This table summarizes	estimation result	ts based on indiv	vidual-level data	h from the $Austr$	ian Birth Regist	er in the period
from 08/1984 through	07/1987. The	dep. var. is eq	ual to one if th	le birth weight	is lower than 25	500 grams, and
zero ounerwise. Intenio was horn before the C	d ot estimation i hernobyl acciden	s a LFIM. THE CONTRACTION	ontroi conort wa nceived between	as concerveu bet 08/1985 and 0	ween Uo/1904 a 1/1986 and was	uu u <i>i</i> /1903 anu hetween 4 and
9 months post concept	tion at the time	of the accident.	BC II was con	iceived between	02/1986 and 04	4/1986 and was
between 0 and 3 mont	hs post conceptic	on at the time of	f the accident.	3C III was conce	eived after Cher	nobyl (between
05/1986 and $07/1987$).	. Each DiD speci	fication controls	for community.	year and month	fixed-effects. R	tobust standard
errors in parentheses.	*, ** and *** inc	licate statistical	. significance at	the 10% , 5% , ar	id 1% level.	

Table 9: The effect of the accident on the likelihood of low birth weight (< 2500 grams)

	Low e	educated mc	others	Highly	educated m	others
	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47~\mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42~{\rm kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$
BCI	0.006	-0.000	-0.007	-0.004	-0.006	-0.003
	(0.023)	(0.024)	(0.024)	(0.013)	(0.014)	(0.014)
BC II	-0.019	-0.016	-0.024	0.013	0.006	0.008
	(0.039)	(0.040)	(0.040)	(0.021)	(0.022)	(0.022)
BC III	0.002	0.002	-0.008	0.019	0.013	0.017
	(0.045)	(0.047)	(0.048)	(0.023)	(0.024)	(0.025)
T # x BCI	-0.012	-0.004	-0.008	-0.024	-0.050^{**}	-0.071^{**}
	(0.029)	(0.033)	(0.043)	(0.019)	(0.022)	(0.027)
T # x BC II	0.026	0.007	-0.014	0.021	0.025	0.024
	(0.047)	(0.054)	(0.068)	(0.017)	(0.021)	(0.027)
T# x BC III	-0.009	-0.013	0.009	0.008	-0.002	-0.001
	(0.021)	(0.025)	(0.028)	(0.011)	(0.012)	(0.015)
Abortion effect	-0.025	-0.015	-0.017	0.017	0.012	0.011
P-value	0.492	0.688	0.653	0.345	0.501	0.544
Conception effect	0.021	0.017	0.016	0.007	0.007	0.010
P-value	0.488	0.584	0.620	0.643	0.628	0.544
Obs.	25372	23572	22343	70293	64265	60603
Mean	9.868	9.869	9.869	9.897	9.896	9.895
This table summarizes	estimation result	ts based on indiv	vidual-level data	i from the $Austr$	ian Birth Regist	er in the period
from 08/1984 through	07/1987. The e	dep. var. is equ	ual to the Apga	ur score after tei	n minutes, and	zero otherwise.
Method of estimation	is a LPM. The co	ontrol cohort wa	s conceived bety	veen 08/1984 an	107/1985 and $107/1985$	was born betore
une Chernobyl acciden concention at the time	u. DC I was con of the seridant	Icelved Detween BC II was for	uo/1903 auu U neaimed between	1/1300 auu was 02/1086 and 0	Detween 4 and 1/1086 and was	9 monus post between 0 and
3 months post concept	tion at the time	of the accident.	BC III was co	nceived after Ch	iernobyl (betwee	en 05/1986 and
07/1987). Each DiD s _F	ecification contr	ols for communi	ty, conception-y	ear and concepti	ion-month fixed-	effects. Robust
standard errors in pare	entheses. *, ** a	nd *** indicate	statistical signi	icance at the 10	%, 5%, and 1%	level.

Table 10: The effect of the accident on the Apgar score 10

	Low e	educated mc	others	Highly	educated m	others
	$37 \ \rm kBq/m^2$	$42 \ \rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47 \ \rm kBq/m^2$
BC I	-0.001	-0.001	-0.002	-0.001	-0.002	-0.002
	(0.003)	(0.003)	(0.003)	(0.002)	(0.002)	(0.002)
BC II	0.005	0.002	0.001	0.001	0.000	0.001
	(0.006)	(0.006)	(0.006)	(0.003)	(0.003)	(0.003)
BC III	0.006	0.003	0.002	-0.001	-0.001	-0.000
	(0.007)	(0.007)	(0.007)	(0.004)	(0.004)	(0.004)
T#xBCI	-0.001	-0.001	-0.004	-0.004^{*}	-0.004	-0.007*
	(0.004)	(0.004)	(0.005)	(0.002)	(0.003)	(0.004)
T # x BC II	-0.008	-0.013	-0.017	0.002	0.001	0.003^{*}
	(0.007)	(0.009)	(0.011)	(0.002)	(0.003)	(0.002)
T#xBCIII	-0.000	-0.005	-0.007*	-0.000	-0.001	-0.001
	(0.003)	(0.003)	(0.004)	(0.001)	(0.002)	(0.002)
Abortion effect	0.006	0.004	0.003	0.002	0.002	0.002
P-value	0.297	0.521	0.556	0.463	0.542	0.435
Conception effect	0.001	0.001	0.001	-0.001	-0.001	-0.001
P-value	0.676	0.732	0.742	0.548	0.562	0.559
Obs.	25986	24163	22917	71695	65610	61914
Mean	0.993	0.993	0.993	0.995	0.995	0.995
This table summarizes	s estimation resul	ts based on indiv	vidual-level data	from the Austr	ian Birth Regist	er in the period
from 08/1984 through	07/1987. The d	ependent variab	le is equal to o	ne if the child is	still alive 24 hc	ours after birth.
Method of estimation	is a LPM. The contact BC_{1} mass contact	ontrol cohort wa	s conceived betr	ween 08/1984 an 1 /1086 and mag	d 07/1985 and botmon 4 and	was born before 0 months nost
conception at the time	e of the accident	BC_{II} was cor	uniceived between	$1/1990$ and was $02/1986$ and 0_4	1/1986 and was	between 0 and
3 months post concep	tion at the time	of the accident.	BC_{III} was co	nceived after Ch	ernobyl (betwe	= 05/1986 and
07/1987). Each speci	fication controls	for community,	conception-yea	r, and conceptic	m-month fixed-	effects. Robust
standard errors in pair and 1-percent.	centneses. *, ** a	nd *** indicate	statistical sign	incance at the 1	J-percent level,	5-percent level,

Table 11: Probability of being still alive 24 hours after birth

	Low e	educated mc	others	Highly	educated m	others
	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47 \ \rm kBq/m^2$
BC I	-0.000	-0.000	-0.001	-0.001	-0.002	-0.002
	(0.004)	(0.004)	(0.004)	(0.002)	(0.002)	(0.002)
BC II	0.006	0.004	0.001	0.002	0.002	0.002
	(0.008)	(0.008)	(0.008)	(0.004)	(0.004)	(0.004)
BC III	0.008	0.005	0.002	0.001	0.001	0.001
	(0.008)	(0.009)	(0.00)	(0.004)	(0.005)	(0.005)
T#xBCI	-0.001	-0.003	-0.003	-0.004	-0.005*	-0.010^{**}
	(0.005)	(0.005)	(0.007)	(0.003)	(0.003)	(0.004)
T#xBCII	-0.007	-0.013	-0.011	0.002	0.003	0.005^{**}
	(0.008)	(0.010)	(0.011)	(0.003)	(0.003)	(0.002)
T#xBCIII	0.003	-0.001	-0.002	0.001	-0.000	-0.001
	(0.004)	(0.004)	(0.005)	(0.002)	(0.002)	(0.002)
Abortion effect	0.006	0.004	0.002	0.004	0.004	0.004
P-value	0.338	0.527	0.731	0.270	0.274	0.218
Conception effect	0.001	0.001	0.001	-0.001	-0.001	-0.002
P-value	0.769	0.808	0.804	0.570	0.527	0.462
Obs.	25986	24163	22917	71695	65610	61914
Mean	0.990	0.990	0.990	0.992	0.992	0.992
This table summarizes	estimation result	ts based on indiv	vidual-level data	i from the $Austr$	ian Birth Regist	er in the period
from 08/1984 through	07/1987. The c	lependent varial	ble is equal to e	one if the child	is still alive 7 d	ays after birth.
Method of estimation	is a LPM. The co	ontrol cohort wa	s conceived bety	veen 08/1984 an	1000000000000000000000000000000000000	was born before
conception at the time	e of the accident	BC_{II} was cor	uo/1303 auu u iceived between	1/1900 allu was 02/1986 and 0	4/1986 and was	between 0 and
3 months post concep	tion at the time	of the accident.	BC_{III} was co	nceived after Ch	, iernobyl (betwe	= 05/1986 and
07/1987). Each speci	fication controls	for community,	conception-yea	r, and conceptic	on-month fixed-	effects. Robust
standard errors in par and 1-percent.	entneses. ', ' a	na · · · maicate	statistical sign	ncance at the 1	u-percent level,	o-percent level,

Table 12: Probability of being still alive 7 days after birth

	Low 6	educated mc	others	Highly	educated m	others
	$37 \ \rm kBq/m^2$	$42~{\rm kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42 \ \rm kBq/m^2$	$47 \ \rm kBq/m^2$
BCI	-0.000	-0.000	-0.001	-0.002	-0.003	-0.003
	(0.004)	(0.004)	(0.004)	(0.002)	(0.002)	(0.002)
BC II	0.008	0.006	0.004	0.002	0.002	0.001
	(0.008)	(0.008)	(0.008)	(0.004)	(0.004)	(0.004)
BC III	0.008	0.006	0.003	-0.000	-0.001	-0.002
	(0.00)	(0.00)	(0.00)	(0.005)	(0.005)	(0.005)
T#xBCI	-0.003	-0.001	-0.003	-0.004	-0.006*	-0.009**
	(0.005)	(0.006)	(0.007)	(0.003)	(0.003)	(0.004)
T#xBCII	-0.006	-0.011	-0.010	0.002	0.004	0.006^{**}
	(0.008)	(0.010)	(0.011)	(0.003)	(0.003)	(0.002)
T#xBCIII	0.001	-0.003	-0.006	0.002	-0.000	-0.000
	(0.004)	(0.004)	(0.005)	(0.002)	(0.002)	(0.003)
Abortion effect	0.008	0.006	0.005	0.005	0.005	0.005
P-value	0.268	0.396	0.482	0.178	0.186	0.204
Conception effect	0.000	-0.000	-0.000	-0.002	-0.003	-0.003
P-value	0.999	0.964	0.922	0.300	0.255	0.227
Obs.	25986	24163	22917	71695	65610	61914
Mean	0.988	0.989	0.989	0.991	0.991	0.991
This table summarizes	s estimation resul	ts based on indiv	vidual-level data	h from the Austr	ian Birth Regist	er in the period
from $08/1984$ through	07/1987. The d	ependent variab	ole is equal to o	ne if the child is	still alive 1 mo	nth after birth.
Method of estimation	is a LPM. The co	ontrol cohort wa	s conceived bety	ween 08/1984 an	d 07/1985 and	was born before
conception at the time	e of the accident	BC_{II} was con	uo, 1303 auu u nceived between	1/1960 allu was 02/1986 and 04	11986 and was	between 0 and
3 months post concep	tion at the time	of the accident.	BC_{III} was co	nceived after Ch	iernobyl (betwe	= 05/1986 and
07/1987). Each speci	fication controls	for community,	conception-yea	r, and conceptic	m-month fixed-	effects. Robust
standard errors in par and 1-percent.	entheses. *, ** 8	und *** indicate	statistical sign	ficance at the 1	0-percent level,	5-percent level,

Table 13: Probability of being still alive 1 month after birth

	Low 6	educated mc	others	Highly	educated m	others
	$37 \ \mathrm{kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42 \ \rm kBq/m^2$	$47 \ \rm kBq/m^2$
BC I	0.000	0.001	-0.000	-0.003	-0.003	-0.003
	(0.005)	(0.005)	(0.005)	(0.003)	(0.003)	(0.003)
BC II	-0.002	-0.000	-0.002	0.001	0.002	0.001
	(0.010)	(0.010)	(0.010)	(0.005)	(0.005)	(0.005)
BC III	0.006	0.007	0.005	-0.002	-0.001	-0.001
	(0.010)	(0.011)	(0.011)	(0.005)	(0.005)	(0.006)
T#xBCI	-0.004	-0.004	-0.004	-0.003	-0.006	-0.010^{*}
	(0.006)	(0.007)	(0.008)	(0.003)	(0.004)	(0.005)
T # x BC II	0.004	-0.004	-0.003	0.001	0.005	0.006
	(0.009)	(0.011)	(0.013)	(0.004)	(0.004)	(0.004)
T#xBCIII	-0.000	-0.003	-0.004	0.002	-0.001	-0.001
	(0.005)	(0.005)	(0.006)	(0.002)	(0.003)	(0.003)
Abortion effect	-0.002	-0.001	-0.001	0.004	0.005	0.004
P-value	0.805	0.920	0.865	0.267	0.245	0.304
Conception effect	0.008	0.007	0.007	-0.003	-0.003	-0.002
P-value	0.167	0.238	0.252	0.298	0.349	0.398
Obs.	25986	24163	22917	71695	65610	61914
Mean	0.983	0.983	0.983	0.988	0.988	0.988
This table summarizes	estimation resul	ts based on indiv	vidual-level data	i from the $Austr$	ian Birth Regist	er in the period
from $08/1984$ through	1 07/1987. The e	dependent varia	ble is equal to	one if the child	is still alive 1 y	ear after birth.
Method of estimation	is a LPM. The co	ontrol cohort wa	s conceived bety	veen 08/1984 an	$^{-1}$ 07/1985 and $^{-1}$	was born before
une Unernobyl accider concention at the time	II. $D \cup_I$ was concerned for the accident	Icelved between BC_{rr} was cor	uo/1965 ana u nceived hetween	1/1960 and was 02/1986 and 0	between 4 and 4/1986 and was	9 monus post between 0 and
3 months post concept	tion at the time	of the accident.	BC_{III} was co	nceived after Ch	iernobyl (betwee	en 05/1986 and
07/1987). Each speci	fication controls	for community,	conception-yea	r, and conceptic	on-month fixed-	effects. Robust
standard errors in par and 1-percent.	entheses. ^{*, **} ^a	nd ^{***} indicate	statistical sign	ficance at the 1	0-percent level,	5-percent level,
······································						

Table 14: Probability of being still alive 1 year after birth

	Low e	educated mo	others	Highly	educated m	others
	$37 \ \rm kBq/m^2$	$42~\rm kBq/m^2$	$47 \ \rm kBq/m^2$	$37~{\rm kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$
BCI	-0.028	-0.037	-0.045	-0.004	-0.012	-0.012
	(0.031)	(0.030)	(0.031)	(0.019)	(0.020)	(0.020)
BC II	-0.021	-0.029	-0.053	-0.039	-0.040	-0.040
	(0.063)	(0.066)	(0.068)	(0.037)	(0.039)	(0.041)
BC III	-0.084	-0.097	-0.126^{*}	-0.032	-0.042	-0.045
	(0.069)	(0.073)	(0.075)	(0.037)	(0.040)	(0.041)
T # x BCI	-0.028	-0.032	-0.036	0.014	0.011	0.009
	(0.040)	(0.046)	(0.058)	(0.019)	(0.024)	(0.026)
T#xBCII	-0.159^{***}	-0.200***	-0.185^{**}	0.011	-0.024	-0.059
	(0.061)	(0.061)	(0.073)	(0.033)	(0.036)	(0.044)
T# x BC III	-0.020	-0.001	0.002	-0.012	-0.011	-0.024
	(0.034)	(0.040)	(0.049)	(0.016)	(0.020)	(0.022)
Abortion effect	0.007	0.008	-0.008	-0.034	-0.028	-0.028
P-value	0.892	0.891	0.885	0.282	0.398	0.415
Conception effect	-0.063	-0.068	-0.073	0.006	-0.002	-0.005
P-value	0.116	0.093	0.076	0.757	0.929	0.794
Obs.	20464	19010	18003	58175	53254	50286
Mean	0.703	0.705	0.708	0.712	0.713	0.713
This table summarizes from 08/108/ through	estimation result	ts based on individual	vidual-level data	from the Austr be number of c	ian Birth Regist	er in the period
the pivotal child (i.e.	the child assigned	ed to one of the	three birth col	norts). Method	of estimation is	OLS. Mothers
are assigned to the co	introl cohort or	the treatment c	cohort dependin	g on the conce _j	ption date of th	e pivotal child.
The control cohort wa	is conceived betw	$reen \ 08/1984 \ an$	id $07/1985~{\rm and}$	was born before	e the Chernobyl	accident. BC_I
was conceived between	108/1985 and 0	1/1986 and was	s between 4 and	9 months post	conception at	the time of the
accident. BC_{II} was c the time of the accide:	onceived between $mt. BC_{III}$ was c	n 02/1986 and 1 conceived after (04/1986 and we Chernobyl (betw	s between 0 an reen 05/1986 an	.d 3 months pos .d 07/1987). Ea	t conception at ch specification
controls for community	v, conception-yea	r, and conceptio	n-month fixed-e	ffects as well as	mother's age du	mmies. Robust
standard errors in par	entheses. *, ** a	nd *** indicate	statistical signi	ficance at the 1	0-percent level,	5-percent level,
and 1-percent.						

fertility
post-treatment
Mothers'
15:
Table

	Boy	s: low educa	ated	Girl	s: low educa	ated
	$37 \ {\rm kBq/m^2}$	$42 \ \rm kBq/m^2$	$47~\mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42~\rm kBq/m^2$	$47 \ \mathrm{kBq/m^2}$
BCI	0.026	0.021	0.006	-0.121**	-0.134**	-0.141^{**}
	(0.049)	(0.050)	(0.051)	(0.056)	(0.056)	(0.057)
BC II	0.067	0.043	0.001	-0.145^{*}	-0.141	-0.158
	(0.096)	(0.100)	(0.101)	(0.087)	(0.092)	(0.096)
BC III	0.008	-0.016	-0.064	-0.220^{**}	-0.226^{**}	-0.247^{**}
	(0.102)	(0.107)	(0.108)	(0.104)	(0.110)	(0.115)
T # x BCI	0.002	-0.019	0.054	-0.054	-0.032	-0.118
	(0.064)	(0.079)	(0.096)	(0.062)	(0.075)	(0.093)
T # x BC II	-0.197^{*}	-0.315^{***}	-0.281^{***}	-0.128	-0.106	-0.128
	(0.110)	(0.087)	(0.094)	(0.084)	(0.085)	(0.103)
T # x BC III	-0.017	-0.017	-0.016	-0.039	-0.006	-0.022
	(0.049)	(0.058)	(0.072)	(0.047)	(0.055)	(0.065)
Abortion effect	0.041	0.022	-0.005	-0.024	-0.007	-0.017
P-value	0.622	0.795	0.955	0.768	0.933	0.841
Conception effect	-0.059	-0.058	-0.065	-0.075	-0.085	-0.089
P-value	0.332	0.345	0.298	0.184	0.141	0.131
Obs.	10490	9747	9239	9974	9263	8764
Mean	0.690	0.689	0.691	0.718	0.722	0.725
This table summarizes	s estimation result	ts based on indiv	vidual-level data	from the $Austr$	ian Birth Regist	er in the period
from 08/1984 through	1 07/1987. The c	lependent varial	ble is equal to t	he number of c	hildren born to	a mother after
une prvotar cumu (ne. are assimned to the or	une cuud assigne	su vo one or vite the treatment <i>c</i>	chort denendin	rotus). Intentiou « on the concer	u esunauou Ation data of th	e nivotal child
The control cohort we	as conceived betw	veen 08/1984 an	d 07/1985 and	was born before	the Chernobyl	accident. BC_I
was conceived betwee:	n $08/1985$ and 0	1/1986 [°] and was	between 4 and	. 9 months post	conception at	the time of the
accident. BC_{II} was c	conceived between	n $02/1986$ and (04/1986 and wa	s between 0 and	d 3 months pos	t conception at
the time of the accide	int. BC_{III} was c	onceived after (Chernobyl (betw	reen $05/1986$ an	d 07/1987). Ea	ch specification
controls for community standard errors in par	y, conception-yea entheses. *, ** a	r, and conceptio nd *** indicate	m-montn nxed-e statistical signi	ficance at the 1	motner's age du 0-percent level,	5-percent level,

and 1-percent.

Table 16: Low educated mothers' post-treatment fertility

	Boys:	highly edu	cated	Girls:	highly edu	cated
	$37~{\rm kBq/m^2}$	$42~\rm kBq/m^2$	$47 \ \rm kBq/m^2$	$37~{\rm kBq/m^2}$	$42~\rm kBq/m^2$	$47~{\rm kBq/m^2}$
BC I	-0.033	-0.044*	-0.049*	0.028	0.025	0.033
	(0.025)	(0.026)	(0.027)	(0.028)	(0.029)	(0.030)
BC II	-0.111^{**}	-0.118^{**}	-0.125^{**}	0.043	0.046	0.053
	(0.051)	(0.053)	(0.055)	(0.054)	(0.057)	(0.059)
BC III	-0.083	-0.101^{*}	-0.118^{*}	0.028	0.026	0.038
	(0.056)	(0.060)	(0.062)	(0.062)	(0.066)	(0.068)
T # x BCI	0.047^{*}	0.068^{*}	0.085^{**}	-0.015	-0.042	-0.057
	(0.028)	(0.035)	(0.043)	(0.030)	(0.037)	(0.040)
T#xBCII	0.063	0.054	0.026	-0.042	-0.107^{**}	-0.146^{**}
	(0.047)	(0.054)	(0.066)	(0.040)	(0.046)	(0.060)
T#xBCIII	0.002	0.016	0.010	-0.025	-0.035	-0.055
	(0.023)	(0.028)	(0.035)	(0.023)	(0.031)	(0.034)
Abortion effect	-0.079	-0.074	-0.076	0.015	0.021	0.020
P-value	0.071	0.099	0.100	0.755	0.673	0.691
Conception effect	0.029	0.016	0.007	-0.015	-0.019	-0.015
P-value	0.351	0.599	0.833	0.616	0.537	0.641
Obs.	29854	27315	25785	28321	25939	24501
Mean	0.713	0.712	0.713	0.711	0.714	0.714
This table summarized from 08/1984 through the pivotal child (i.e.	i estimation result 07/1987. The c the child assigned	es based on indi- lependent varia ed to one of the	vidual-level data ble is equal to 1 e three birth col	the from the $Austrice$ the number of control of the north of the nor	<i>ian Birth Regist</i> hildren born to of estimation is	er in the period a mother after s OLS. Mothers
are assigned to the control other control other control other control other control co	ontrol cohort or 1	the treatment of	cohort dependin	g on the concer	otion date of th	e pivotal child.
the control contort wi was conceived betwee	as concerved between $08/1985$ and 01	een 00/1904 an 1/1986 and was	id Ut/1965 and s between 4 and	Was DOFIL DELOFE 9 months post	conception at	accident. $D \cup_I$ the time of the
accident. BC_{II} was o	conceived between	$n^{'}02/1986$ and	04/1986 and we	is between 0 and	d 3 months pos	t conception at
the time of the accide controls for communit	put. BC_{III} was c y, conception-yeau	onceived after (r, and conceptic	Chernobyl (betv n-month fixed-€	reen 05/1986 an ffects as well as	d 07/1987). Ea mother's age du	ch specification immies. Robust
standard errors in par and 1-percent.	entheses. *, ** a	nd *** indicate	statistical sign	ficance at the 1	0-percent level,	5-percent level,

fertility
post-treatment
mothers'
educated
Highly
Table 17:

	Boy	s: low educa	ated	Girl	s: low educ:	ated
	$37 \ \rm kBq/m^2$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$	$37~{\rm kBq/m^2}$	$42 \ \rm kBq/m^2$	$47 \ \rm kBq/m^2$
BCI	0.033	0.015	0.008	0.009	0.025	0.022
	(0.053)	(0.053)	(0.054)	(0.053)	(0.055)	(0.057)
BC II	0.104	0.082	0.064	0.039	0.065	0.046
	(0.069)	(0.071)	(0.073)	(0.073)	(0.074)	(0.076)
BC III	0.091	0.074	0.043	0.065	0.095	0.081
	(0.087)	(0.091)	(0.094)	(0.086)	(0.088)	(0.091)
T # x BCI	0.008	-0.015	-0.060	-0.017	0.006	0.040
	(0.038)	(0.041)	(0.047)	(0.031)	(0.032)	(0.039)
T # x BC II	-0.118^{**}	-0.129^{**}	-0.125^{*}	0.038	0.041	0.017
	(0.054)	(0.059)	(0.070)	(0.044)	(0.052)	(0.063)
T # x BC III	0.032	0.010	0.010	-0.013	-0.003	-0.025
	(0.025)	(0.027)	(0.034)	(0.029)	(0.032)	(0.039)
Obs.	9943	9226	8741	9431	8755	8273
Mean	0.529	0.528	0.530	0.514	0.516	0.516

Table 18: Low educated mothers' labor force participation eight years after childbirth

dependent variable is equal to one if the mother of the pivotal child (i.e. the child assigned to one of the three was conceived between 08/1985 and 01/1986 and was between 4 and 9 months post conception at the time of the This table summarizes estimation results based on individual-level data from the Austrian Social Security Data. The birth cohorts) participates in the labor force eight years after childbirth. Method of estimation is LPM. Mothers the time of the accident. BC_{III} was conceived after Chernobyl (between 05/1986 and 07/1987). Each specification The control cohort was conceived between 08/1984 and 07/1985 and was born before the Chernobyl accident. BC_I accident. BC_{II} was conceived between 02/1986 and 04/1986 and was between 0 and 3 months post conception at are assigned to the control cohort or the treatment cohort depending on the conception date of the pivotal child. controls for community, conception-year, and conception-month fixed-effects as well as mother's age dummies. Robust standard errors in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent.

	Boys:	highly edu	cated	Girls:	highly edu	cated
	$37 \ \mathrm{kBq/m^2}$	$42 \ \mathrm{kBq/m^2}$	$47 \ \mathrm{kBq/m^2}$	$37 \ \rm kBq/m^2$	$42~\rm kBq/m^2$	47 kBq/m^2
BCI	0.025	0.016	0.016	0.017	0.017	0.026
	(0.026)	(0.027)	(0.028)	(0.030)	(0.032)	(0.033)
BC II	0.065^{*}	0.049	0.040	0.002	-0.006	-0.001
	(0.034)	(0.035)	(0.036)	(0.041)	(0.043)	(0.043)
BC III	0.065	0.044	0.032	-0.004	-0.022	-0.010
	(0.043)	(0.044)	(0.045)	(0.048)	(0.049)	(0.050)
T # x BCI	0.019	0.022	0.043	-0.000	0.003	-0.019
	(0.021)	(0.028)	(0.035)	(0.018)	(0.023)	(0.026)
T # x BC II	-0.039^{*}	-0.053*	-0.061^{*}	-0.011	-0.052	-0.051
	(0.023)	(0.029)	(0.035)	(0.036)	(0.035)	(0.040)
T # x BC III	0.006	0.009	0.013	0.004	0.011	-0.007
	(0.014)	(0.018)	(0.021)	(0.014)	(0.017)	(0.021)
Obs.	29011	26538	25047	27512	25210	23813
Mean	0.570	0.570	0.571	0.569	0.569	0.570

$\operatorname{childbirth}$
after
years
eight
ipation
partic
force
labor
mothers'
educated
Highly
9 19:
Table

This table summarizes estimation results based on individual-level data from the Austrian Social Security Data. The dependent variable is equal to one if the mother of the pivotal child (i.e. the child assigned to one of the three birth cohorts) participates in the labor force eight years after childbirth. Method of estimation is LPM. Mothers was conceived between 08/1985 and 01/1986 and was between 4 and 9 months post conception at the time of the the time of the accident. BC_{III} was conceived after Chernobyl (between 05/1986 and 07/1987). Each specification accident. BC_{II} was conceived between 02/1986 and 04/1986 and was between 0 and 3 months post conception at are assigned to the control cohort or the treatment cohort depending on the conception date of the pivotal child. The control cohort was conceived between 08/1984 and 07/1985 and was born before the Chernobyl accident. BC_I controls for community, conception-year, and conception-month fixed-effects as well as mother's age dummies. Robust standard errors in parentheses. *, ** and *** indicate statistical significance at the 10-percent level, 5-percent level, and 1-percent.

Appendix A Existing evidence on the effects of the Chernobyl accident on reproductive outcomes

The International Commission on Radiological Protection (ICRP) considers an effective dose of 100 mSv as a threshold for effects after in utero exposure to ionizing radiation, including the induction of cancer. Whether the Chernobyl accident caused any negative health effects on individuals living in European countries is still under debate, despite the much lower radiation dose those individuals were exposed to. (See for example the recent debate in The Lancet (Holt, 2010).)

The existing evidence on the health impact of in utero exposure to the Chernobyl accident (summarized below) is mainly based on epidemiological studies analyzing time trends (across differently exposed regions¹) in the rates of live births, stillbirths, spontaneous (and induced) abortions, infant mortality and perinatal and postnatal outcomes (e.g. pre-term birth, low birth weight, congenital malformations, incidence of specific diseases). Any (short-term) deviations from the long-run trend after the Chernobyl accident are cautiously interpreted as evidence for radiation-related health effects. However, most authors acknowledge the limited power of their studies to detect small effects and emphasize that causal inference is hardly possible based on ecological studies. In light of these limitations, most reviews of the existing evidence conclude that there is no consistent evidence of detrimental effects of the Chernobyl disaster except for an increase in thyroid cancer for individuals exposed in childhood, particularly in Belarus, Ukraine and Russia. (e.g. Little, 1993; WHO, 2006; UNSCEAR, 2000).

There is some evidence for an increase in the proportion of stillbirths and the early infant (or perinatal) mortality rate after the Chernobyl accident in Germany.² However, these results have been challenged by other studies using German data and studies for other countries (e.g. Finland, Sweden).³ Furthermore, there is no evidence for a significant relationship between the level of fallout and the rate of spontaneous abortions, congenital malformations or other postnatal outcomes (pre-term birth, low birth weight, childhood cancer).⁴ In contrast, there is some evidence for a decrease in the birth rate (independent of the fallout level) 9-11 months after the accident in Sweden, Finland, Norway and Italy and a temporary increase in the rate of induced abortions in Greece, Italy and Sweden.⁵

¹However, this is not done in a difference-in-difference framework.

²See Lüning *et al.* (1989); Scherb *et al.* (1999); Körblein and Küchenhoff (1997); Scherb *et al.* (2000). ³See Blettner (2000); Grosche *et al.* (1997); Auvinen *et al.* (2001); Ericson and Kallen (1994).

⁴See Auvinen *et al.* (2001); Ericson and Kallen (1994); Irgens *et al.* (1991); Haeusler *et al.* (1992); Harjulehto *et al.* (1989).

⁵See Auvinen *et al.* (2001); Ericson and Kallen (1994); Bertollini *et al.* (1990); Irgens *et al.* (1991); Trichopoulos *et al.* (1987). Haeusler *et al.* (1992) find no effect on the counseling rate at pregnancy termination clinics in southern Austria.

Both effects may be attributed to the conflicting information in the media and the anxiety of pregnant women in the first month after the Chernobyl accident.